Author’s response to reviews

Title: Strengthening health data on a rare and heterogeneous disease: Sarcoma incidence and histological subtypes in Germany

Authors:
Meike Ressing (ressing@uni-mainz.de)
Eva Wardelmann (eva.wardelmann@ukmuenster.de)
Peter Hohenberger (peter.hohenberger@umm.de)
Jens Jakob (jens.jakob@umm.de)
Bernd Kasper (bernd.kasper@umm.de)
Katharina Emrich (emrich@krebsregister-rlp.de)
Andrea Eberle (eberle@leibniz-bips.de)
Maria Blettner (blettner-sekretariat@uni-mainz.de)
Sylke Zeissig (zeissig@krebsregister-rlp.de; zeissig@uni-mainz.de)

Version: 2 Date: 25 Jan 2018

Author’s response to reviews:
Sarcoma incidence and histological subtypes in Germany* for publishing in BMC Public Health. Please find below our response to your comments:

1. Please ensure that all additional files are cited in the text; currently additional table 2 is not cited.

It is now cited in line 204.

2. Please ensure that all authors are referenced in the 'Authors' Contributions' declaration, currently AE and AAB are missing. Please also ensure contributions are reflective of ICMJE guidelines (http://www.icmje.org/recommendations/browse/roles-and-responsibilities/defining-the-role-of-authors-and-contributors.html).
In the sentence “All authors contributed to the interpretation of data, revised the manuscript carefully and approved its final version.” in the authors contributions statement, we replaced „All authors“ by „EW, PH, JJ, BK, KE, AE, MB and SRZ“. There is no author with the initials AAB in this manuscript.

3. At this stage, please upload a single, clean revised version of the manuscript (including clean versions of all relevant tables/figures) that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.

We cleaned the manuscript and all tables and figures.

Kind regards
Sylke Zeissig